Author's response to reviews

Title: Determinants of malnutrition in children and the uptake of interventions to prevent the condition

Authors:

Edem M.A Tette (edemenator@googlemail.com)
Eric K Sifah (ericsifah@yahoo.com)
Edmund T Nartey (etnartey@chs.edu.gh)

Version: 2
Date: 6 May 2015

Author's response to reviews: see over
Authors response to comments

Determinants of malnutrition in children and the uptake of interventions to prevent the condition

1. Ethics:

Please note that all submissions to BMC Pediatrics must comply with our editorial policies. In particular, please ensure that your manuscript contains any required statements regarding ethical approval and consent from patients/participants; adheres to any relevant standards of reporting with a completed checklist where appropriate; and contains an Availability of Supporting Data section explaining how the supporting data is available to researchers, including the repository and accession number where appropriate. Manuscripts which do not adhere to our editorial policies may experience delays or be rejected. Further information about our editorial policies can be found at the following links:

Response:

The following statement has been inserted into manuscript (Line 122-127): Ethical approval was sought and obtained from the University of Ghana Medical School’s Ethical and Protocol Review Committee [Protocol Identification Number: MS-Et/M.8-P.5.8/2011-2012]. Ethical approval was also obtained from the Ghana Health Service Ethical Review Committee [Protocol Identification Number GHS-ERC 05/07/2012]. Written consent was obtained from the study participants using consent forms which were signed or thumb printed.